Author’s response to reviews

Title: Sodium valproate induced acute pancreatitis in a bipolar disorder patient: a case report

Authors:

Wanli Huang (pungpungli@outlook.com)

Xin Ren (monster_831@163.com)

Fang Shen (jek1998@126.com)

Baoping Xing (xbp2003@hotmail.com)

Version: 4 Date: 13 Nov 2019

Author’s response to reviews:

Dear Editors,

Thank you for your mail concerning our manuscript entitled “Sodium valproate induced acute pancreatitis in a bipolar disorder patient: a case report” (manuscript number: PHAT-D-18-00290R3). The main corrections in the paper and the responds to the editor’ comments are as followed:

1. On the Title Page, the full names of all authors are used.
2. In the ‘Funding’ section, the role of the funding body is addressed.
3. We confirm a copy of the consent for publication form is available for review by the Editor of the journal, and the statement is included.
4. Files we do no wish to be published have been removed.
5. Our manuscript, a final and clean version, have been uploaded.

We tried our best to improve the manuscript and made some changes in it. These changes will not influence the content and framework of the paper. We appreciate for Editors’ warm work earnestly, and hope that the correction will meet with approval. Once again, thank you very much for the comments and suggestions.